Author's response to reviews

Title: Exploring the Usefulness of Comprehensive Care Plans for Children with Medical Complexity (CMC): A Qualitative Study

Authors:

Sherri Adams (sherri.adams@sickkids.ca)
Eyal Cohen (eyal.cohen@sickkids.ca)
Sanjay Mahant (sanjay.mahant@sickkids.ca)
Jeremy N Friedman (jeremy.friedman@sickkids.ca)
Radha MacCulloch (radha.macculloch@mail.mcgill.ca)
David B Nicholas (nicholas@ucalgary.ca)

Version: 2 Date: 31 August 2012

Author's response to reviews: see over
August 31st, 2012

Re: Exploring the Usefulness of Comprehensive Care Plans for Children with Medical Complexity (CMC)

To the Editorial Board of BMC Pediatrics,

Thank you kindly for considering this manuscript for publication in BMC Pediatrics. I am writing to address the concern below:

Editor Comment: Consent statement: Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

New line: page 6. “Written, informed consent was obtained from all parent and HCP participants prior to their participation in the study.”

Please contact me with any additional questions or concerns.

Sincerely,

Sherri Adams